Author's response to reviews

Title: Variable viral clearance despite adequate ganciclovir plasma levels during valganciclovir treatment for cytomegalovirus disease in D+/R- transplant recipients

Authors:

Nancy Perrottet (Nancy.Perrottet@chuv.ch)
Oriol Manuel (oriol.manuel@chuv.ch)
Frédéric Lamoth (frederic.lamoth@chuv.ch)
Jean-Pierre Venetz (jean-pierre.venetz@chuv.ch)
Roland Sahli (roland.sahli@chuv.ch)
Laurent A Decosterd (laurentarthur.decosterd@chuv.ch)
Thierry Buclin (thierry.buclin@chuv.ch)
Manuel Pascual (manuel.pascual@chuv.ch)
Pascal Meylan (pascal.meylan@chuv.ch)

Version: 6 Date: 30 December 2009

Author's response to reviews:

A few typos were corrected